Author's response to reviews

**Title:** Influence of social and material individual and area deprivation on suicide mortality among 2.7 million Canadians: A prospective study

**Authors:**

Stephanie Burrows (burrows.stephanie@sympatico.ca)
Nathalie Auger (nathalie.auger@inspq.qc.ca)
Philippe Gamache (philippe.gamache@inspq.qc.ca)
Danielle St-Laurent (danielle.st.laurent@inspq.qc.ca)
Denis Hamel (denis.hamel@inspq.qc.ca)

**Version:** 6 **Date:** 15 July 2011

Author's response to reviews:

The Editorial Production Team
BMC Public Health Editorial Office

15 July 2011

Dear Editorial Production Team

RE: MS: 1882397776508576 - Influence of social and material individual and area deprivation on suicide mortality among 2.7 million Canadians: A prospective study.

Thank you for your careful reading of our manuscript. As requested, we have made the following formatting changes to the manuscript:

1. Tables 3a, 3b, 4a and 4b have been renamed as Table 3, Table 4, Table 5 and Table 6. The reference to them in the text has also been changed (Page 10 last paragraph, page 11 first paragraph).
2. All highlighting/tracking has been removed from the manuscript.
3. The manuscript has been checked for any typographical errors.

Please do no hesitate to contact me should there be any queries or additional changes required.

Sincerely yours,

Stephanie Burrows